Author's response to reviews

Title: Comparative evaluation of diode laser versus argon laser photocoagulation in patients with central serous retinopathy: A prospective, randomized clinical trial

Authors:
Lalit Verma (lalitverma@yahoo.com)
Rajesh Sinha (sinharaj1@rediffmail.com)
Pradeep Venkatesh (venkyprao@yahoo.com)
H K Tewari (hktewari1@yahoo.com)

Version: 3 Date: 22 October 2004

Author's response to reviews: see over
Dr Christopher Gadd
Assistant Editor
BMC Journals

**Re:** Submission of the revised manuscript no MS: 1153700457370693, entitled “Comparative evaluation of diode laser versus argon laser photocoagulation in patients with central serous retinopathy: A prospective, randomized clinical trial”

Dear Dr. Gadd,

Enclosed please find the revised copy of the above mentioned manuscript for consideration for publication in your journal. The article has been formatted as per the format of the journal and as per your suggestions.

I hope the paper now fulfils the requirements for publication in your esteemed journal.

Thanking you

With warm regards

Yours sincerely,

Rajesh Sinha
Formatting changes in the manuscript

- **ISRCTN** – The ISRCTN has been incorporated into the title of the manuscript.
- **Change tracking** – As suggested, we have turned off all change tracking.
- **Text** - The text 'Presented in part in the Annual Meeting of the American Academy of Ophthalmology, 1997 in San Francisco, California' has been moved to the last sentence of the manuscript.
- **Competing interests** – As suggested, this statement has been removed from the title page and has been shifted in the end.
- **Section headings** - The colons from the ends of the section headings in the abstract have been removed.
- **Material and Method** – It has been renamed as ‘Methods’.
- **References** - References have been cited in the text using consecutive numbers in square brackets.
- **References** - In the manuscript text, the reference has been placed before the full stop.
- **Individual contribution of authors** – It has been renamed as ‘Authors’ contributions’.
- **References** – All authors have been mentioned in the Reference section as suggested and the suffix ‘et al’ has been removed.
- **Tables** - Table 1a and 1b have been renamed as Table 1 & 2.
- **Figure legends** – The figure legends have been moved to after the reference list.
- **Figures** – The figures have been uploaded as separate files.